Author’s response to reviews

Title: Longitudinal Study of Electrical, Functional and Structural Remodelling in an Equine Model of Atrial Fibrillation

Authors:
Eva Hesselkilde (evah@sund.ku.dk)
Helena Carstensen (hc@sund.ku.dk)
Mette Flethøj (flethoj@sund.ku.dk)
Merle Fenner (merle.fenner@sund.ku.dk)
Ditte Kruse (ditte.kruse@sund.ku.dk)
Stefan Sattler (sattler@sund.ku.dk)
Jacob Tfelt-Hansen (jacob.tfelt@regionh.dk)
Steen Pehrson (steen.pehrson@regionh.dk)
Thomas Hartig Braunstein (thobra@sund.ku.dk)
Jonas Carlson (jonas.carlson@med.lu.se)
Pyotr Platonov (pyotr.platonov@med.lu.se)
Thomas Jespersen (thojes@sund.ku.dk)
Rikke Buhl (rib@sund.ku.dk)

Version: 2 Date: 26 Jul 2019

Author’s response to reviews:

Dear Editor, dear reviewer 1 and reviewer 2,

Please find enclosed our revised manuscript. We are very grateful for your thorough revision of our manuscript. Your detailed comments and suggestions were of great benefit for the improvement of the manuscript.
We respect your comments and suggestions in almost every detail and are very glad to hear, that our work was favourably received and you see merit to our study.

As requested, the revised manuscript (using the track changes mode in MS Word) is accompanied by a point-by-point response (in italic) to every comment made by you.